# Design and Test of an Optoelectronic-VLSI Chip With 540-Element Receiver–Transmitter Arrays Using Differential Optical Signaling

Michael B. Venditti, Associate Member, IEEE, Emmanuelle Laprise, Julien Faucher, Member, IEEE, Pierre-Olivier Laprise, J. Eduardo A. Lugo, and David V. Plant, Member, IEEE

Abstract—We have constructed an optoelectronic very-largescale integration (OE-VLSI) chip with a 540-element receiver and transmitter array. Differential optical signaling was used in conjunction with a fully differential electrical architecture for the receiver and transmitter circuits. The chip was partitioned into multiple functional channels to demonstrate different chip-to-chip communication functions appropriate for applications of OE-VLSI technology. Wide optical input-output busses were provided for each channel in order to demonstrate high degrees of parallelism. The architecture and design of the chip are described in detail, including the digital functionality, the optoelectronic devices, and the arrays of receiver and transmitter circuits. The design verification of the chip is also described. We present experimental results that both verify the full functionality of the chip design and verify that the receiver and transmitter circuits and digital circuitry met their designed performance targets.

Index Terms—Application specific integrated circuits, differential optical signaling, driver circuits, mixed analog–digital integrated circuits, optical interconnections, optical receivers, optical trasmitters, optoelectronic very-large-scale integration (OE-VLSI), very-large-scale integration (VLSI).

# I. INTRODUCTION

**T** WO-DIMENSIONAL (2-D) optical data links (ODLs) are capable of delivering the required connectivity and aggregate bandwidths between elements in computing and switching systems [1]–[4]. The 2-D format of this interconnect technology makes it inherently scalable. Optoelectronic very-large-scale integration (OE-VLSI) circuits that combine the processing power of silicon complementary metal–oxide–semiconductor (CMOS) with the efficiency of GaAs-based emitters, such as

Manuscript received October 18, 2002; revised February 3, 2003. This work was supported in part by BAE Systems under a contract via the DARPA/ARL VLSI-Photonics program (DAAL01-98-C-0074), in part by the Canadian Microelectronics Corporation (CMC), and in part by the Canadian Institute for Photonic Innovation (CIPI). The work of M. Venditti and J. Faucher was supported by the National Sciences and Engineering Research Council (NSERC) of Canada. The work of J. Faucher and E. Laprise was supported by Le Fonds pour la Formation de Chercheurs et l'Aide à la Recherche (FCAR) du Québec. The work of J. Faucher was also supported by McGill University under an R. H. Tomlinson doctoral fellowship.

M. B. Venditti and E. Laprise were with the Photonics Systems Group, Department of Electrical and Computer Engineering, McGill University, Montreal, QC H3A 2A7, Canada. They are now with PMC-Sierra, Montreal, QC H3R 3L5, Canada (e-mail: Michael\_Venditti@pmc-sierra.com).

J. Faucher, P.-O. Laprise, J. E. A. Lugo, and D. V. Plant are with the Photonics Systems Group, Department of Electrical and Computer Engineering, McGill University, Montreal, QC H3A 2A7, Canada.

Digital Object Identifier 10.1109/JSTQE.2003.812512

vertical-cavity surface-emitting lasers (VCSELs) and photodetectors (PDs), represent an enabling technology [5]–[8].

Using this technology, we have constructed an OE-VLSI chip with 1080 PDs and 1080 VCSELs that employs fully differential optical signaling to realize 540-element receiver and transmitter arrays that provide optical input–output (I/O) to the chip in addition to conventional electrical I/O. The PD and VCSEL arrays were heterogeneously integrated onto a CMOS VLSI chip using a flip-chip bonding and substrate removal process similar to that described elsewhere [9]. In addition to implementing multiple digital functions suitable for OE-VLSI technology, the receiver, transmitter, and digital circuits were designed for robustness, testability, and ease of operability.

Several large-scale OE-VLSI chips previously reported have employed a mix of single-ended [9], [10] and differential [11]–[14] optical signaling. With the exception of a small subset of critical ODLs in [10], however, all of the underlying electrical architectures of the receiver and transmitter circuits in these chips have been single-ended. For those chips that employed differential optical signaling, a principal motivation was to overcome the poor contrast ratio available from the use of multiple quantum-well devices in modulator-based transmitters.

The chip presented in this work employs differential optical signaling in conjunction with a fully differential electrical architecture for the receiver and transmitter circuits with the intention of improving the operability of large arrays of receivers and transmitters. Fully differential electrical architectures provide greater immunity to the effects of crosstalk and power supply switching noise [15], [16]. The use of differential optical signaling with a dc-coupled fully differential receiver allows the receiver to perform self-threshold decision making through common-mode rejection [15]. This avoids the need to implement an offset control function to overcome the operational problems of large arrays of single-ended receivers that have a fixed decision threshold [17], [18]. In single-ended optical signaling, switching noise that affects the output of the transmitter is optically transmitted to a receiver and directly impairs its operation, even when a differential electrical receiver architecture is employed. The use of differential optical signaling provides greater immunity for a receiver to optically transmitted switching noise [18], [19].

The drawbacks of using differential optical signaling are that the usable ODL density is halved as compared with single-ended signaling, and transmitter power dissipation is

1077-260X/03\$17.00 © 2003 IEEE



Fig. 1. Photograph of the chip after heterogeneous OED integration, highlighting its principal sections: receiver array, digital section, and transmitter array.

larger. In large-scale OE-VLSI applications, where the number of available ODLs is on the order of hundreds or thousands, halving the usable ODL density may not be a major sacrifice given the potential benefits described. Although the transmitter power dissipation increases with differential optical signaling due to the need to bias twice as many lasers, the additional power penalty diminishes as lasers with smaller threshold currents are developed. For tight optoelectronic device (OED) pitches of, for example, 125  $\mu$ m, the physical space available to implement a receiver or transmitter circuit with single-ended optical signaling can severely limit the circuit complexity that can be implemented. Thus, from a physical implementation perspective, differential optical signaling can be advantageous as it provides twice as much area to implement the receiver or transmitter.

This chip is to be used in a system demonstrator that implements a point-to-point interchip link using an optical system based on free-space optical interconnects or on fiber image guides. The objective of the system is to demonstrate different functions appropriate for applications of OE-VLSI technology in chip-to-chip interconnects. The chip was partitioned into four separate functional channels, with optical I/O busses provided for each channel and a single, shared electrical I/O bus whose width (128 bits) matches the needs of the optical I/O busses.

This paper discusses the architecture and design of the chip in detail, including its digital functionality, the OEDs, and the receiver and transmitter circuits and arrays. The design verification of the chip is also described. Experimental verification of the chip performance and design is described in detail, including the development of the printed circuit board packaging and a custom software interface that was used to control the operation of the chip. Although the printed circuit board (PCB) used for the chip testing reported in this paper had limitations that prevented exhaustive optical testing of large numbers of ODLs simultaneously, it was possible to verify the performance of the chip at the targeted data rates.

The paper is organized as follows. Section II presents the CMOS chip architecture. Section III describes the digital section of the chip, consisting mainly of the digital-functional channels. Section IV describes the construction of the receiver and transmitter arrays, including the OEDs and heterogeneous integration, and the receiver and transmitter circuit design. Section V describes the multistep design verification of the chip, including the design of the validation PCB, the software interface, and the test benches that were developed. Section VI presents the experimental results obtained. Section VII is a summary and discussion section.

# II. CMOS CHIP ARCHITECTURE

Multiple digital-functional channels were implemented on the chip to demonstrate applications suitable for OE-VLSI technology. The digital-functional channels included a mix of bit-serial processing/routing (often referred to as "smart-pixel" applications) [9], [14], [20]-[24], and bit-parallel processing functions [25], [26]. Given a channel-based approach, a modular chip floor plan [25] was chosen. A photograph of the chip after heterogeneous OED integration is shown in Fig. 1, with the principal sections of the chip indicated: the receiver array, the digital section, and the transmitter array. These sections are connected with highly parallel on-chip electrical interconnects. The digital-functional blocks, to be described in subsequent sections, were arranged in the middle section of the chip, with electrical I/O, control, and power connections distributed along the top and bottom in multiple rows of flip-chip (C4) bonding pads. The receiver and transmitter arrays were placed on the left and right



Fig. 2. Block diagram of the ASIC. The middle portion shows the digital section of chip, and the left- and right-hand portions show the receiver and transmitter arrays, respectively.

sides of the digital section of the chip, respectively, with dedicated differential pairs of optical inputs and outputs for each of the four digital channels. Power connections were placed along the top and bottom sections of the receiver and transmitter arrays in multiple rows of C4 pads. Some electrical I/O and control pads were placed on the left side of the receiver array and on the right side of the transmitter array. The chip had a total of 879 off-chip electrical connections, including 47 for serial data, control, and clocking, 128 each for the electrical input and output busses, 32 for digital power and ground, and 272 each for power and ground connections for the receiver and transmitter arrays. Multiple supply voltages were provided for power for the receivers (2.5 V), the transmitters (3.3 V), the digital section (2.5 V), and the electrical I/O pads (3.3 V). All of these supply voltages were independent of one another on-chip.

The receiver and transmitter circuits were designed for operation at 250 Mb/s to match the performance of the electrical I/O pads and most of the digital-functional channels. Achieving the target performance for the receiver design was complicated by the need for the receiver to drive long interconnect busses out of the receiver array. Achieving the target performance for the transmitter design was simpler since the long interconnect busses into the transmitter array were driven by the digital circuitry.

The 14.6 by 7.5-mm chip was designed in a 0.25- $\mu$ m five metal, single poly, n-well CMOS process, and fabricated by

the Taiwan Semiconductor Manufacturing Company. The Canadian Microelectronics Corporation<sup>1</sup> provided access to this technology through the MOSIS service.<sup>2</sup> Arrays of VCSELs and PDs were heterogeneously integrated with the chip to provide optical I/O capability. Access to heterogeneous OED integration via flip-chip bonding was provided by BAE Systems.<sup>3</sup> In Sections III and IV, we describe the design of the digital section of the chip and the design of the receiver and transmitter arrays in detail.

#### **III. DIGITAL SECTION FUNCTIONALITY**

The middle portion of Fig. 2 shows a block diagram of the digital section of the chip. The chip was partitioned into four separate functional channels, each of which is described in Sections III-A–D. Each channel was designed using conventional digital design methods. The Verilog hardware description language (HDL) [27] was used for coding the functionality of each channel, and Synopsys' Design Compiler software was used to synthesize these designs to the gate level. Cadence Design Systems' Design Planner and Silicon Ensemble software were used

<sup>&</sup>lt;sup>1</sup>Canadian Microelectronics Corporation, Kingston, ON, Canada [Online]. Available: www.cmc.ca.

<sup>&</sup>lt;sup>2</sup>The MOSIS service, Marina del Rey, CA, USA [Online]. Available: www.mosis.org.

<sup>&</sup>lt;sup>3</sup>BAE Systems—South Nashua Facility, Nashua, NH, USA [Online]. Available: www.baesystems.com.

to perform automatic placement and routing for some channels and channel subblocks. Cadence Design Systems' IC Craftsman software was used to interconnect the subblocks and channels at the top level.

Banks of centrally located multiplexer circuits were used to connect the electrical input and output busses to the digital channels. The electrical input bus could be connected to multiple channels simultaneously. Only one channel (the active channel) could access the electrical output bus at any time.

Multiple sets of input and output buffers were used between the input and output bus pads, the digital channels, and the centrally located multiplexer circuits. Special care was taken in matching trace lengths and matching the number of buffers used between the different bits of the electrical busses in order to reduce skew.

The control register is a register file containing thirteen 8-bit registers that control the function and monitor the status of the digital channels. The control register is also used to set the connectivity of the electrical input and output busses. Individual registers in the register file are accessed by address. Writing to and reading from a register is performed bit-serially. To reduce the noise induced on the clock lines by the data input lines (i.e., crosstalk), a finite state machine (FSM) was implemented to control the programming function of the control register. In order to modify the contents of a register, a specific 4-bit sequence had to be input to the FSM.

The receiver and transmitter arrays served as optical inputs and outputs for the digital channels. Electrical data from the receiver array were buffered before being accessed by the digital channels. The transmitter design required complementary data inputs. Thus, the complement of each signal destined for the transmitter array was obtained using an inverter, and then both the data and its complement were buffered before being sent to the transmitter array. Care was taken in the selection of the inverter and buffer circuits to ensure that skew was not introduced in the complementary transmitter inputs.

# A. First-In First-Out Channel

A typical problem encountered in OE-VLSI application-specified integrated circuit (ASIC) design is that the performance of the optical interconnect generally exceeds that of electrical I/O pads [28], [29]. In applications where data on a wide data bus are obtained from an off-chip source to be transmitted optically, the performance of the electrical interconnect can be a bottleneck. The first-in first-out (FIFO) channel (ffChan) allows the optical transfer of data between two chips while decoupling the performance of the optical interconnect from that of the electrical I/O busses. Electrical data can be loaded slowly from an off-chip source and then burst-transmitted optically. The channel consists of a send FIFO and a receive FIFO, each implemented as a 128-bit by 16-word dual-port static random access memory (SRAM) block. Independent read and write clocks are used to load data to and read data from each FIFO.

Data transfer between two chips is illustrated in Fig. 3. Fig. 3(a) shows the flow diagram of the *send* FIFO on the first chip. The 16 128-bit data words from the electrical input bus



Fig. 3. Simplified flow diagram of data transfer between two chips using the *ffChan*. (a) *Send* FIFO on first chip. (b) *Receive* FIFO on second chip.

are written into the *send* FIFO using a write clock. An electrical read clock (operated by the read controller) is then used to read out the 16 128-bits data words from the *send* FIFO and send them to the transmitter array for optical transmission to the second chip. The electrical read clock is also sent to the transmitter array and, with a four-fold redundancy, is transmitted optically to the second chip as a strobe signal. Fig. 3(b) shows the flow diagram of the *receive* FIFO on the second chip. The received strobe signals from the first chip are demultiplexed, and the optimal signal is selected and used as a write clock to load the 16 128-bit data words received from the first chip into the *receive* FIFO. An electrical read clock is used to read out the 16 128-bit data words from the *receive* FIFO and send them to the electrical output bus.

The *send* and *receive* FIFOs in the *ffChan* can operate independently; thus, it is possible to establish bidirectional point-to-point communication between two chips. It is also possible to operate a single chip in an optical loop-back configuration, where the outputs of the transmitter array are connected to the inputs of the receiver array.

Features were added to the *ffChan* to facilitate the electrical-only testing of the *send* and *receive* FIFOs. The 128-bit serial scan chain (SSC) registers were added before the input port and after the output port of both the *send* and *receive* FIFOs, allowing the electrical input and output busses to be bypassed. An additional test feature (not shown in Fig. 3) was added where the output of the *send* FIFO could be connected directly to the input of the *receive* FIFO, bypassing the transmitter and receiver arrays. Using these test features, it was possible to test the read and write functions of each FIFO without relying on optical I/O.

#### B. Data Generation Channel

The data generation channel (*dgChan*) was implemented to facilitate the bit-error testing of large transmitter and receiver arrays at high speeds. It incorporates a 16-bit linear feedback shift register (LFSR) that generates a  $2^{16}-1$  bit pseudorandom bit sequence (PRBS). The *dgChan* consists of 128 corresponding sets



Fig. 4. Simplified flow diagram of the *dgChan*. (a) *Toggle/scan* cell. (b) *Comparator* cell.

of *toggle/scan* and *comparator* cells (described below), transmitter and receiver circuits, and electrical input and output bus bits.

Simplified flow diagrams for the toggle/scan and comparator cells are shown in Fig. 4. Fig. 4(a) shows the flow diagram of a toggle/scan cell, which can operate in toggle or scan mode. All 128 toggle/scan cells form an SSC, where the Scan Out pin of one cell in the chain was connected to the Scan In pin of the next cell in the chain. An enable signal is provided for each toggle/scan cell from a corresponding bit on the electrical input bus. If the enable bit is held low, the corresponding transmitter circuit is sent Logic 0. In toggle mode, the toggle/scan cell generates a square wave at one half the frequency of the input clock. In scan mode, the first toggle/scan cell in the scan chain takes its input from the output of the LFSR, and the PRBS is shifted through the chain of *toggle/scan* cells, optically transmitting the PRBS data. Although this approach allows for the generation of a large amount of PRBS data, the data appearing at neighboring toggle/scan cells are correlated in time [30].

Fig. 4(b) shows the flow diagram of a *comparator* cell. Each *comparator* cell could be used to perform rudimentary error detection by comparing optically received data against the expected data stream generated by the LFSR. The PRBS data from the corresponding *toggle/scan* cell is used for comparison against data obtained from the corresponding receiver circuit. If the received data is correct, the *comparator* cell output is Logic 1; otherwise, it is Logic 0. The corresponding bit on the electrical output bus can either be the output of the *comparator* cell or the direct output of its corresponding receiver.

Optical data transfer can take place between the transmitters and receivers of the same chip or between two different chips. In the single-chip case, the *toggle/scan* cells, *comparator* cells, and the LFSR operate from the same clock signal, making synchronization simple. In the two-chip case, more elaborate synchronization is required. The *master* chip performs the PRBS data generation and optical transmission. The *slave* chip receives optical data and performs the comparison. The *slave* chip also generates the PRBS data in phase with the *master* chip using its local LFSR by synchronizing its clock with that of the *master* chip and also by receiving a "start" signal from it. This is accomplished using clock and enable signals transmitted optically from the *master* chip to the *slave* chip. As these two signals are essential to the operation of the *dgChan*, the master chip optically transmits them each with four-fold redundancy. The *slave* chip subsequently demultiplexes these signals.

Features were added to the *dgChan* to facilitate the electrical-only testing of the *comparator* and *toggle/scan* cells. The D flip-flops (DFFs) within the *comparator* cells were connected in an SSC. Both this SSC and the one comprised of DFFs within the *toggle/scan* cells were accessible directly via chip I/O pads. They could be used to bypass the transmitter and receiver arrays as well as the electrical output bus. Using these test features, it was possible to test the read and write functions of each FIFO without relying on optical I/O.

## C. Feed-Through Channel

The feed-through channel (*ftChan*) was implemented to perform data routing functions similar to those implemented in other smart-pixel based OE-VLSI chips [9], [14], [20]–[24]. It is capable of performing *add*, *drop*, and *feed-through* functions on 128 bits of data. An *ftChan* unit cell consisted of a receiver circuit from the receiver array, a multiplexer, a corresponding transmitter circuit from the transmitter array, and corresponding bits on the electrical input and output busses. In the *add* mode, data from the electrical input bus are transmitted optically by the transmitters. In the *drop* mode, optical data are received by the receivers and placed on the electrical output bus. In the *feed-though* mode, optical data are receivers and immediately retransmitted optically by the transmitters as well as being placed on the electrical output bus.

Unlike the other digital channels, the *ftChan* does not provide any specific electrical-only test features. The *ftChan* is, however, the digital channel that provides the most direct access to transmitter and receiver circuits, without the need for any clocking. Many of the test results presented in Section VI were obtained using the *ftChan*.

# D. Error Correction Channel

In the early development of OE-VLSI chips and systems, attaining high reliability of the optical interconnect has been a principal yet difficult goal to attain. Yield problems beyond those normally associated with VLSI fabrication can result in an ODL having a higher than normal bit-error rate (BER) or one that is permanently inoperative. Such problems can occur, for example, during heterogeneous OED integration or during system packaging and alignment with an optical system, which can result in dead VCSELs, dead PDs, or reduced optical link power throughput due to aberrations or misalignment in the optical system.

The use of forward error correction (FEC) techniques in long-haul optical communication systems is a common approach to improving reliability and is typically based on long block lengths and high information rates using time-sequential (i.e., bit-serial) encoding and decoding functions [31], as illustrated in Fig. 5(a). The use of data (de)multiplexing in the time domain is also common in such schemes. In OE-VLSI systems, the density of the optical interconnect precludes the need for data (de)multiplexing. It is also preferable to avoid time-sequential encoding and decoding because a permanently inoperative ODL cannot be overcome. Fig. 5(b) illustrates an



Fig. 5. Comparative flow diagrams of FEC implementations. (a) Time-sequential (bit-serial) FEC approach with a serial data stream and a serial codec. (b) Parallel FEC approach using a parallel codec to transmit all bits of a data word in parallel.

FEC approach using a parallel decoder that receives an entire code word each clock cycle from the parallel transmission of optical data on multiple ODLs [32]–[34]. In such an approach using an appropriate FEC scheme, it is possible for one or more permanently inoperative ODLs to exist and still maintain error-free communication.

The error correction channel (*ecChan*) was implemented as a parallel coder–decoder pair (codec). The FEC scheme selected was the Golay code [35]. The block length of the Golay code is 24 bits, its information rate is 0.5 (it is thus referred to as a (24, 12) code), and it is capable of correcting up to three errors and detecting up to four errors in a 24-bit encoded word. The selected implementation of the (24, 12) Golay codec, which uses purely combinational circuitry, is particularly suitable for VLSI implementation due to its compactness and low latency [32], [36].

The *ecChan* has six subchannels, each consisting of 24 receiver circuits, 24 transmitter circuits, an encoder, a decoder, and the 12 corresponding payload bits on the electrical input and output busses. In total, the *ecChan* uses 72 bits of data on the electrical input and output busses and 144 receiver and transmitter circuits. Although twice as many ODLs are required to achieve the same throughput of an unencoded link for the same ODL data rate, it should be noted that, theoretically, the use of parallel FEC could achieve an improved ODL data rate due to coding gain [32].

There are three modes of operation for the *ecChan*. In the *normal* mode, there are only combinational circuit elements in the data paths. The six subchannel encoders directly encode the 72 bits of data on the electrical input bus, with the resulting 144 bits of encoded data sent to the transmitter circuits. On the decoding side, the 144 bits of data obtained by the receiver circuits are decoded by the six subchannel decoders into 72 bits and placed on the electrical output bus. In the *test* mode, a number of sequential circuit elements are added to the data paths to facilitate electrical-only testing. The encoder has SSC registers at its inputs and outputs that can be used to bypass the electrical input bus and the transmitter circuits, respectively. The decoder also has SSC registers at its inputs and outputs that can be used

to bypass the receiver circuits and the electrical output bus, respectively. The decoder has additional registers to allow errors to be inserted into the decoder input and to monitor the internal behavior of the decoder. In the *bypass* mode, the encoder and decoder of each subchannel are bypassed. No data processing is performed in this mode, facilitating the testing of the remaining circuitry.

#### IV. RECEIVER AND TRANSMITTER ARRAYS

The left- and right-hand portions of Fig. 2 show block diagrams of the receiver and transmitter arrays, respectively. The receiver and transmitter arrays are comprised of 540 receiver and 540 transmitter circuits and serve as optical I/O for the digital-functional channels described in Section III. The receiver and transmitter circuits in the arrays [discussed in Sections IV-A-D] were organized into groups that were controlled independently to allow for operational flexibility. Each channel had 128 bits of optical I/O organized as four common-control groups of 32 bits in each array, as indicated in Fig. 2. Additional optical I/O was allotted to some channels on the right-hand side of the receiver array and the left-hand side of the transmitter array, as indicated in Fig. 2. The ffChan had an additional 4 bits of optical I/O used for the read and write clocks for the channel's send and receive FIFOs, respectively. The dgChan had an additional 8 bits of optical I/O used for clock and enable signals. The ecChan had two additional sets of 8 bits of optical I/O. These additional sets of bits for the ecChan are located adjacent to the 32-bit common-control groups of the ftChan and ecChan in the receiver and transmitter arrays (refer to Fig. 2). All additional sets of transmitter and receiver circuits formed independent common-control groups in each array.

A vertical column of digital buffers was inserted approximately in the middle of the receiver and transmitter arrays, as shown in Fig. 2. This was done to break up the long electrical interconnects out of the receiver array and into the transmitter array into shorter and more uniform segments. The control registers for the receiver and transmitter arrays were located at the vertical center of their respective arrays, with some dedicated I/O and power pads on the left-hand side of the receiver array and the right-hand side of the transmitter array, as shown in Fig. 2. These control registers were used to digitally set the magnitudes of various parameters (feedback resistance magnitude for the receivers and bias and modulation current magnitude for the transmitters) and to control the enable and test signals for each common-control group. The power pads along the top of the arrays provide power for the receiver and transmitter circuits for the ffChan and dgChan, and the power pads along the bottom of the arrays provide power for the receiver and transmitter circuits for the *ftChan* and *ecChan*. The supply voltage for the receiver array was 2.5 V for voltage-level compatibility with the digital section of the chip, to which it directly interfaces. The supply voltage for the transmitter array was 3.3 V to accommodate the relatively large forward-bias voltage drop of the VCSELs. The inputs to the transmitter array are generated directly from the digital section of the chip and are rail-to-rail 2.5 V CMOS signals.

The receiver and transmitter circuits were designed for operation at a data rate of 250 Mb/s. Sections IV-A–D describe the characteristics and modeling of the PDs and VCSELs used in the receiver and transmitter arrays, the design of the receiver and transmitter circuits, and the formation of the receiver and transmitter arrays.

# A. PD and VCSEL Properties, Modeling, and Integration

The 34-row by 35-column PD and VCSEL arrays were heterogeneously integrated with the CMOS chip using two flip-chip bonding and substrate removal procedures similar to that described in [9]. Of the 1190 devices in each OED array, 1080 were used with transmitter or receiver circuits. The remaining 110 OEDs in each array are physically present but not electrically connected to receiver or transmitter circuits. This occurs in the two horizontal rows of OEDs that covered the receiver and transmitter array control registers and the vertical column of OEDs that covered the buffers in the receiver and transmitter arrays (refer to Fig. 2). Additionally, two sets of four OEDs near the four-element common-control group of the *ffChan* (refer to Fig. 2) were not connected to receiver or transmitter circuits.

The PDs were fabricated by BAE Systems with square active areas of 50  $\mu$ m on a side and 15 ×15  $\mu$ m pads for the p and n contacts. The array was constructed as a tile of two-row by one-column unit cells on 125- $\mu$ m horizontal and 250- $\mu$ m vertical pitches. Within a unit cell, the p and n contacts of the top and bottom PDs were mirrored about the horizontal axis. The PD active areas were placed on a 125- $\mu$ m vertical pitch. When the unit cells were tiled, the active areas of all PDs were on a 125- $\mu$ m grid. Fig. 6(a) and (b) shows photographs of a 2 × 2 PD subarray before and after heterogeneous integration, respectively. Unit cells are indicated. This specialized PD contact arrangement was beneficial for routing receiver outputs out of the receiver array and is discussed in greater detail in Section IV-D.

The VCSELs were fabricated by EMCore Corp.<sup>4</sup> with 10- $\mu$ m diameter active areas and 15 × 15  $\mu$ m pads for the p and n con-

<sup>4</sup>EMCore Corp., Somerset, NJ, USA [Online]. Available: www.emcore.com.



Fig. 6. Photographs of a  $2 \times 2$  PD subarray (a) before and (b) after heterogeneous integration with the chip. p and n contacts and a unit cell are indicated. PD active areas are on a 125- $\mu$ m horizontal and vertical pitch. Unit cells are on a 125- $\mu$ m horizontal and 250- $\mu$ m vertical pitch.



Fig. 7. Photographs of a  $2 \times 2$  PD subarray (a) before and (b) after heterogeneous integration with the chip. The p and n contacts are indicated. The p and n contacts and VCSEL active areas are on a 125- $\mu$ m horizontal and vertical pitch.

 TABLE I

 PD AND VCSEL OPTICAL AND ELECTRICAL PROPERTIES

PD Parameter	Symbol	Value
responsivity	R	$0.317 \text{ A/W} \pm 10.8 \text{ mA/W}$
junction capacitance	$C_J$	500 fF
dark current	I <sub>dark</sub>	6 nA
VCSEL Parameter	Symbol	Value
threshold current (@25°C)	$I_{TH} _{25^{\circ}C}$	$1.400 \text{ mA} \pm 75 \mu\text{A}$
slope efficiency (@25°C)	$\eta _{25^{\circ}C}$	$0.340 \text{ W/A} \pm 10 \text{ mW/A}$
$I_{\rm TH}$ temperature dependence	$\partial I_{TH} / \partial T$	15.1 μA/°C
$\eta$ temperature dependence	$\frac{\partial \eta}{\partial T}$	-2.2 mW/A·°C
differential resistance (2-5 mA)	$\partial R _{2-5mA}$	$71.7\Omega\pm12.1\Omega$
threshold voltage (@ 2mA)	$V_{TH} _{2mA}$	$1.580 \text{ V} \pm 17 \text{ mV}$
junction capacitance	$C_J$	500 fF

tacts and were arranged on a standard 125- $\mu$ m grid. Fig. 7(a) and (b) shows photographs of a 2 × 2 VCSEL subarray before and after heterogeneous integration, respectively.

Measurements were taken on OED samples to determine their nominal optical and electrical properties and their sensitivity to temperature variations. Table I summarizes the optical and electrical properties of the samples. All property values are measured quantities except for those stated as exact numbers, which are quoted from OED manufacturer data sheets. The PD responsivity across the sample array was found to be highly uniform and insensitive to reverse bias voltage, temperature, wavelength, and incident power. Most VCSEL parameters were also found to be fairly uniform across the sample array, with the differential resistance being an exception.

During receiver design, the PDs were modeled using lumped circuit elements, with optical power represented as a voltage signal. A capacitor was used to model the junction capacitance, and was placed in parallel with a voltage controlled current source to model the conversion of incident optical power to input photocurrent, using R as the transconductance gain parameter. Two different models were used for the VCSELs during transmitter design. One was a simple model using lumped circuit elements. A dc voltage source  $V_{TH}$  was placed in series with a resistor  $\partial R$ , and both elements were placed in parallel with a capacitor  $C_J$  to model the junction capacitance. The VCSEL output power, represented as a voltage signal, was modeled using a current-controlled voltage source using  $\eta$  as the transresistance gain parameter. Although satisfactory for initial transmitter designs and used frequently in the literature [8], [9], [37], this lumped element model was unable to model critical elements of VCSEL operation such as below- or near-threshold operation and the temperature dependencies of  $I_{TH}$  and  $\eta$ . A more accurate VCSEL model based on behavioral modeling [38]–[40] was developed using Verilog-A, a behavioral HDL for analog circuits. The Verilog-A model was based on the lumped circuit model, but it incorporated the typical diode-like exponential current-voltage relationship, as well as the temperature dependence of the threshold current and slope efficiency, as detailed in Table I.

## B. Receiver Circuit Design

One of the principal reasons that differential optical signaling was employed was to accommodate the dc-coupled nature of the input photocurrent. Optically single-ended receivers have a fixed decision threshold, and variations in the average input photocurrent across an array of receivers can cause severe operational problems in receiver groups that are commonly biased and/or controlled [17]. In optically differential receivers, a fully differential preamplifier architecture with common-mode feedback (CMFB) circuitry [41] stabilizes the operating point and common-mode output voltage of the preamplifier in the face of variations in common-mode input photocurrent.

Two preamplifier designs were implemented. One is based on a feedback-free common-gate amplifier (CGA) with a diode-connected load, and the other is based on a conventional differential transimpedance amplifier (TIA) with resistive feedback. Transistor-level schematics of the CGA and TIA are shown in Fig. 8(a) and (b), respectively. A common feature of the two preamplifier designs was the inclusion of circuitry (transistors MP in Fig. 8) to allow for functional circuit testing prior to heterogeneous OED integration. After heterogeneous integration, the PDs would appear in parallel with transistors MP, as indicated by the dashed lines in Fig. 8. The corresponding active-low digital control inputs  $nV_{\rm tl}$  and  $nV_{\rm tr}$  for the receivers in all common-control groups that form a digital-functional channel are connected together, and can be used to inject small amounts of current (approximately 60  $\mu$ A) into either input of the preamplifier circuit, mimicking a differential input photocurrent.

The feedback resistances used in the TIA preamplifier configuration [resistors RF in Fig. 8(b)] are implemented using active devices and can be tuned using digital control inputs [42]. The transistor-level implementation of the feedback resistance is shown in Fig. 9. One reference N-type metal-oxide-semiconductor (NMOS) transistor (MR) and four other NMOS transistors (M0-M3) are all connected in parallel. MR was kept permanently conducting by having its gate terminal connected to the receiver supply voltage. This establishes a nominal resistance equal to approximately 16 k  $\cdot \Omega$  for small current magnitudes. Transistors M0 through M3 have width-to-length (W/L) ratios progressively increasing by a factor of two. M0 is the smallest with the same W/L ratio as MR. When M0-M3 are made conductive by setting their corresponding gate terminal control voltages  $(V_{\rm ctrl}(0-3))$  to a digital Logic 1 voltage, resistor values of approximately 16, 8, 4, and 2 k  $\cdot \Omega$ , respectively, are established. A total of 16 different effective RF magnitudes can be established from 16 down to 1 k  $\cdot \Omega$ . The active-high  $V_{\text{ctrl}}\langle 0-3 \rangle$ control inputs are common to all receivers in a common-control group, with different control input sets for each common-control group.

Fig. 10 shows a block diagram of the receiver and is applicable for either preamplifier design. There are four circuit stages that follow the preamplifier, including two postamplifier stages, a Schmitt-Trigger inverter stage, and a line driver stage. The two postamplifier stages are parallel NMOS- and P-type metal-oxide-semiconductor-based folded cascode differential amplifiers that employ feedback to maintain bias point stability [43]. They are intended to amplify the preamplifier output to signal levels that approach the receiver power supply rails and to convert the differential signal into a single-ended one. The hysterysis provided in the transfer characteristic of the Schmitt-Trigger [44] stage provides immunity to switching noise on the receiver power supply rails [9], [45]. The line driver stage is a pair of cascaded inverters with a W/L stage ratio of approximately three. Its purpose is to ensure rail-to-rail receiver operation and to drive the on-chip interconnect to the nearest buffer, which can be up to 2 mm away, at the target data rate of 250 Mb/s.

A representative receiver layout with the CGA-based preamplifier design is shown in Fig. 11. The receiver layout for the TIA-based preamplifier design is very similar. The layout is divided into two separate regions, each of which is protected by guard rings to immunize the circuit from substrate noise [46]. The line driver stage is in the smaller region on the right side of Fig. 11, isolated from all of the other receiver circuit stages in the larger region on the left. Isolating the line driver stage helps prevent the switching noise it generates from affecting the operation of the other receiver circuit stages and neighboring receivers. The region with the line driver stage has  $25-\mu$ m horizontal and  $18-\mu$ m vertical dimensions. The region with the remaining receiver circuit stages has approximately  $80-\mu$ m horizontal and  $36-\mu$ m vertical dimensions. The two layout regions



Fig. 8. Transistor-level schematics for the (a) CGA and (b) TIA preamplifier circuits illustrating CMFB, test mode circuitry, and control inputs. The locations of the PDs after heterogeneous OED integration are indicated by dashed lines.



Fig. 9. Transistor-level schematic of TIA preamplifier feedback resistance RF.



Fig. 10. Receiver block diagram, applicable to either preamplifier design.



Fig. 11. Layout of receiver with CGA-based preamplifier design. The line driver stage (right-hand side region) was isolated from the other circuit stages to mitigate switching noise effects.

are separated by approximately 25  $\mu$ m. This separation could have been made larger to improve isolation, but would have required the Schmitt–Trigger stage to be designed to drive a longer interconnect, which would have resulted in additional switching noise being generated.

The power dissipation of the receiver circuit was estimated to be between 8.5 and 9.5 mW per receiver under most operating conditions at a data rate of 250 Mb/s. The *ftChan*, which uses the smallest number of receivers, would dissipate approximately

Fig. 12. Simplified transistor-level schematic of transmitter circuit, illustrating test mode circuitry and control inputs. The location of the VCSELs after heterogeneous OED integration are indicated by dashed lines.

1.23 W. The *ecChan*, which uses the largest number of receivers, would dissipate approximately 1.37 W.

#### C. Transmitter Circuit Design

The optically differential transmitter circuit is based on a current steering design, and its simplified schematic is shown in Fig. 12. A current steering design was chosen over a current switching design [30] to minimize generation of power supply switching noise. Each VCSEL is offset biased with a bias current (IB). The intermediate modulation current (IM) was steered through one of the two VCSELs by transistors M1L and M1R using complementary rail-to-rail CMOS inputs  $V_{IN}$  and  $V_{INB}$ . Operation of the transmitter is as follows: In the logic low (high) transmitter state with  $V_{IN}$  low (high) and  $V_{INB}$  high (low), IM is steered through M1R (M1L), causing  $P_{OB}(P_O)$  to output a large amount of optical power and  $P_O(P_{OB})$  to output a smaller amount of optical power.

The IM and IB current sources shown in Fig. 12 are tunable from control circuits with digital control inputs called the modulation and bias current control block (MCCB and BCCB), respectively [42]. The MCCB and BCCB are digital to analog converters and are identical except for the sizes of their constituent transistors; those in the MCCB are twice as large as their BCCB counterparts. The inputs to the MCCB and BCCB are sets of five active-low digital control signals used to encode the current magnitude. For the MCCB, the range of settable IM currents was from 0 to 4.96 mA in increments of 160  $\mu$ A. For the BCCB, the range of settable IB currents was from 0 to 2.48 mA in increments of 80  $\mu$ A. Fig. 13 shows the simplified schematic of the MCCB and BCCB. Each digital control input is connected to a corresponding switchable current source. The least significant control bit  $nV_{\text{ctrl}}\langle 0 \rangle$  is connected to the smallest current source, and subsequent control bits  $nV_{\text{ctrl}}\langle 1-4\rangle$  are connected to current sources whose magnitudes progressively increase by a factor of two. A diode-connected transistor collects current from the activated current sources and generates the control signal  $V_G$ . Each common-control transmitter group has its own MCCB and BCCB, and the generated  $V_G$  signals are used to set the magnitudes of IM and IB for all transmitter circuits in the



Fig. 13. Simplified schematic of the MCCB and BCCB. All current sources and transistors in the MCCB are twice as large as their counterparts in the BCCB.



Fig. 14. Transmitter circuit layout.

common-control group. Each common-control group of transmitter circuits has independent  $nV_{ctrl}\langle 0-4\rangle$  control inputs.

The transmitter design includes circuitry to allow for circuit testing prior to heterogeneous OED integration in the form of transistors M2L and M2R (see Fig. 12). These additional transistors would be in parallel with the VCSELs after heterogeneous OED integration is performed, as indicated by the dashed lines in Fig. 12. The corresponding active-low digital control inputs  $nV_{tl}$  and  $nV_{tr}$  are common to all transmitters in all common-control circuit groups in a digital-functional channel. When enabled, M2L and M2R allow electrical paths to exist for the IM and IB currents to flow when the VCSELs are not present in the circuit topology.

The physical layout of the transmitter is shown in Fig. 14. The layout utilizes guard rings to immunize the circuit from switching noise via the substrate. The layout area has dimensions of 150  $\mu$ m horizontally and 55  $\mu$ m vertically.

The power dissipation of the transmitter circuit is  $VDD \cdot (2 \times IB + IM)$  and is largely dependent on the set magnitudes for IM and IB. Under worst-case conditions of IB = 2.48 mA and IM = 4.96 mA, power dissipation would be as high as 32.7 mW per transmitter. For more typical operating conditions of IB = 2 mA and IM = 4 mA, power dissipation per transmitter would be 26.4 mA. The transmitters for an entire digital-functional channel would dissipate between 4.2 and 4.7 W of power in the worst case and between 3.4 and 3.8 W of power in the more typical case.

# D. Array Construction

The construction of the receiver and transmitter array layouts was performed manually using a bottom-up approach. Receiver unit cell layouts for each of the two preamplifier designs and the transmitter unit cell layout were prepared, including vertically oriented power and ground rails and appropriately located contact pads and passivation windows for OED integration. Upon completion, the unit cell receiver and transmitter layout blocks were easily tiled to form the common-control circuit groups for the digital-functional channels. In the case of the receivers, a common-control group always consisted of receiver designs of the same type—CGA and TIA designs were never mixed.

Upon completion of the layouts for the common-control receiver and transmitter circuit groups, the layouts of the portions of the receiver and transmitter array for each of the digital-functional channels was prepared by tiling the layouts of the common-control circuit groups. With the exception of the *ftChan*, all common-control receiver groups that form a digital-functional channel consisted of receiver designs of the same type to maintain uniform receiver latency within a channel. The common-control groups of the *ffChan* and *ecChan* used TIAbased receiver designs, whereas those of the *dgChan* used CGAbased receiver designs. In the case of the *ftChan*, uniform latency was not a design consideration, and the receiver designs used for the four common-control groups were evenly split between TIA- and CGA-based designs.

The interconnections between the receiver outputs and buffer circuits and between the buffer circuits and transmitter inputs were then routed. The layouts of the two receivers in a receiver unit cell were mirrored about the center horizontal axis of the unit cell to match the specialized PD contact arrangement described earlier in Section IV-A. In the receiver design, the p and n contacts of each PD were connected to the preamplifier input and to the receiver supply voltage, respectively. The supply voltages were routed vertically using the top two metal layers. Thus, by performing the horizontal mirroring of the PD contacts and receiver layouts, wide horizontal channels devoid of any interfering circuitry or lower level metal layers were created below the PD n contacts when receiver unit cells were tiled. These wide routing channels facilitated the routing of the many receiver outputs out of the receiver array with increased spacing, helping to reduce electrical crosstalk. Fig. 15 illustrates a  $2 \times 2$  layout tile of receiver unit cells, including the locations of passivation windows for the PD p and n contacts. The locations of the routing channels are indicated.

Completion of the receiver and transmitter arrays was performed by positioning each section of the array corresponding to the digital-functional channels, adding the layouts for the power pads, buffer circuits, and the receiver control I/O and then manually routing all the electrical interconnections.

# V. POSTFABRICATION ASIC DESIGN VERIFICATION

# A. Validation PCB

A custom PCB was designed and fabricated to assist in design verification and to perform testing of the ASIC. The backside of the validation PCB contained a conductive area



Fig. 15. Representative  $2 \times 2$  layout tile of receiver unit cells. The receiver layout of a unit cell is symmetric about the horizontal. The unit cell tile results in wide horizontal routing channels for the receiver outputs out of the receiver array.

for packaging the ASIC using a chip-on-board approach. Capacitors were placed around the ASIC to decouple the power supplies and reduce switching noise. The conductive area was plated with gold, contained thermal vias to aid in the distribution of heat dissipated by the ASIC and was maintained at ground potential. Due to wirebond pitch limitations on the chip and the validation PCB, bonding fingers were provided to connect to only a subset of the pads on the ASIC. These included the rows of pads that provide power to the receiver and transmitter arrays that are closest to the chip edges and one row of power pads for the digital section of the chip and for the chip pads themselves. Additionally, bonding fingers were provided for all of the scan chain I/O, control, and clock pads for the digital section of the chip and the receiver and transmitter control registers. There were no connections for the electrical input and output busses. Of the 879 chip pads, only 216 wirebond connections were made to bonding fingers of the validation PCB. Fig. 16(a) shows a photograph of the portion of the backside of the validation PCB, where the ASIC is packaged. Fig. 16(b) shows a photograph of the entire backside of the validation PCB. It should be noted that the ASIC was designed for packaging via flip-chip bonding to a PCB or fanout substrate for the incorporation of the chip in a point-to-point interchip link.

The front side of the validation PCB contains all of the components used to operate the chip, and is shown in Fig. 16(c). Connectors provide raw power to sets of voltage regulators for the digital section of the chip and the top and bottom portions of the receiver and transmitter arrays. The voltage regulators serve to minimize resistive voltage drops on these power supply lines. A 100-pin connector was used to interface the various control



(a)



Fig. 16. Photographs of chip packaging and validation PCB. (a) Close-up of packaged and wirebonded chip. (b) Backside of validation PCB. (c) Front side of validation PCB.

and data lines on the chip with a digital I/O card inside a computer running a custom software interface program described in Section V-B. This ASIC-computer interface is capable of operating only at slow speeds on the order of 100 kHz. To allow the chip to be operated significantly faster, one of the global clock signals on the chip was connected to a high-speed connector on the validation PCB. The high-speed connector was positioned close to the corresponding clock pad of the chip.

#### B. Software Interface

A custom software interface was developed to control a National Instruments PCI-DIO96 digital I/O card. The I/O card interfaced with and controlled all operational aspects of the ASIC on the validation PCB. A graphical user interface (GUI) was used to control low-level capabilities, including the ability to write or read individual pins on the ASIC and to write or read any of the registers in the digital section control register and the receiver and transmitter control registers. Additional GUIs, which make use of the low-level routines, were developed to facilitate control of the receiver and transmitter arrays when performing optical testing in the laboratory.

Higher level test bench routines and GUIs were developed to perform extensive testing of the digital-functional channels and control registers. These test benches were ported to the C++ programming language directly from the Verilog test bench routines used to verify the design of the ASIC prior to fabrication. Several scan tests were implemented for the control registers, the SSCs at the input and output ports of the ffChan send and receive FIFOs, the SSCs for the toggle/scan and *comparator* cells of the *dgChan*, and the SSCs at the inputs and outputs of the encoders and decoders of the ecChan. These scan tests scan a series of test vectors through the SSCs and verifies that the same vectors are scanned out. Test vector generation could be achieved either deterministically or randomly. For the ffChan, a number of memory tests were implemented to detect defects in the FIFO SRAM storage elements. These tests write vectors to the SRAM, and then read them back out for verification. Multiple sequence types could be generated to test for various defect classes [47], such as random vectors, constant-increment vector sequences, and walking zeroes (01111, 10111, 11011, 11101, 11110, ...) and ones  $(10000, 01000, 00100, 00010, 00001, \ldots)$ . For the *ec*-Chan, special test benches were developed to test the encoding and decoding functions. Unencoded vectors could be scanned in to test the encoder. Pre-encoded vectors could be scanned in to test the decoder. The pre-encoded vectors could also be

scanned in and have errors injected prior to being decoded to test the error-correction capabilities of the decoder.

# C. Design Verification

Postfabrication verification of the chip design was performed in multiple steps. During fabrication of the chip, the software interface was debugged using a simulated version of the ASIC [48]. Verilog and Verilog-A HDL models were developed for the digital circuitry and the receiver and transmitter arrays. The complete HDL model for the chip was simulated within the Affirma software environment from Cadence Design Systems, running on a SUN Ultra-60 workstation. The software interface for the chip, running on an IBM-compatible personal computer, interfaced with the simulated chip by replacing the PCI-DIO96 card interface code with function calls from the Verilog programming language interface, allowing the I/O pads of the simulated chip to be accessed through TCP/IP ports.

Upon completion of chip fabrication and software interface debugging, a chip, without heterogeneous OED integration having been performed, was packaged on a validation PCB. Electrical-only testing was performed using the software interface to verify the chip design by running through the suite of digital test benches described in Section V-B. During initial testing, a design error was discovered where a clock and a data input pad were incorrectly wired to the internal circuitry. These errors were corrected on a number of fabricated chips using a focused ion beam microsurgery technique from Fibics, Inc.<sup>5</sup> The repaired chips subsequently passed all of the digital test benches. Electrical-only testing was also performed on the receiver and transmitter circuits, making use of the  $nV_{\rm tl}$ and  $nV_{\rm tr}$  digital control inputs, as discussed in Section IV. For the receivers, these control inputs were used to force all outputs to a common logic state, which were checked using the available SSC registers in the digital-functional channels. For the transmitters, an indirect form of operational testing was used. The transmitter inputs were accessed via the available SSC registers in the digital-functional channels. Using these inputs, along with the  $nV_{\rm tl}$  and  $nV_{\rm tr}$  control inputs and the IB and IM magnitude control inputs, various current paths in the transmitter were established and eliminated, allowing crude testing to be performed by monitoring changes in the current drawn by the voltage regulators on the validation PCB.

# VI. EXPERIMENTAL TEST RESULTS

After the ASIC design was verified, heterogeneous OED integration was performed on additional chips which were subsequently packaged on validation PCBs. Most of the test benches described in Section V were run on these chips to ensure they were free from manufacturing defects. Subsequently, a number of optical experiments were performed to characterize the postintegration qualities of the receiver and transmitter circuits.

The postintegration yield of the VCSELs in the transmitter array was determined by setting the bias currents of all the transmitters to forward-bias all of the VCSELs in the array. There

<sup>5</sup>Fibics, Inc., Ottawa, ON, Canada [Online]. Available: www.fibics.com.

Fig. 17. All VCSELs in transmitter array biased below threshold. Inoperative VCSELs at any bias current indicated by white dots. Dark regions correspond to VCSELs not connected to transmitter circuits.



Fig. 18. Average light–current characteristics for transmitters in a common-control group of the dgChan. The IB/64 curve was obtained by sweeping IB with IM held at 0. The IM/32 curve was obtained by sweeping IM with IB held at 0.

were 1047 of 1080 VCSELs operative, for a yield of 96.9%. Fig. 17 shows the illuminated transmitter array biased below threshold. The inoperative VCSELs are indicated by white circles and were found to occur in a mix of random and clustered locations. It should be noted that the dark regions (one single vertical column, two horizontal rows, and two sets of  $2 \times 2$  squares in the top-left section) of the otherwise illuminated array correspond to the locations of VCSELs that are not connected to transmitter circuits, as described in Section IV-A.

The light–current characteristics for the VCSELs in a representative common-control group (the common-control group of the *dgChan* in which all 64 VCSELs are operative) are shown in Fig. 18. The IB/64 curve was obtained by measuring the aggregate output power of all 64 VCSELs as the IB magnitude setting was swept through all 32 possible states from 0 to 2.48 mA with IM kept at 0 mA. The IM/32 curve was obtained by setting all of the transmitters in the group to be in Logic State 0 and measuring the aggregate output power of 32 VCSELs as the IM magnitude setting was swept through all 32 possible states from



Fig. 19. Average VCSEL output power for IB = 2 mA in each common-control group. The spatial arrangement of common-control groups is a horizontally mirrored version of the one shown on the right-hand side of Fig. 8. For each common-control group, aggregate power was measured and divided by the number of operative VCSELs.

0 to 4.96 mA with IB kept at 0 mA. For VCSEL current between 0 and 2.5 mA, the average  $I_{TH}$  was 1.605 and 1.571 mA, and the average  $\eta$  was 0.184 and 0.190 W/A for the IB/64 and IM/32 curves, respectively.

Comparing these results to Table I, we find that the experimentally measured  $I_{TH}$  is slightly larger than the specified value of 1.40 mA, whereas  $\eta$  is significantly smaller than the 0.340-W/A specification. An increasingly elevated VCSEL operating temperature as the bias and modulation currents are increased is one factor in the discrepancy because there was no means available to stabilize the temperature of the chip. Based on the threshold current temperature coefficient from Table I, the elevated  $I_{TH}$  obtained experimentally corresponds to an elevated temperature of approximately 12.5 °C, assuming no other factors contributed to the increase. Using a similar analysis with the slope efficiency temperature coefficient, the reduced  $\eta$  corresponds to an elevated temperature of approximately 70 °C, assuming no other factors contributed to the decrease. Clearly, temperature was not the sole factor that resulted in an elevated  $I_{TH}$  and a reduced  $\eta$ . Experimental uncertainty in performing the measurements is also a factor. Any presence of  $I_{TH}$  and  $\eta$ variations for individual VCSELs within the common-control transmitter group were subject to averaging effects that could distort the results. Additionally, imperfect current mirroring in the MCCB and BCCB could result in either a larger or smaller IB or IM current flowing in the VCSELs than the intended settings. It is also possible that the VCSEL properties were degraded as a result of the heterogeneous integration process.

The VCSEL output power in each common-control transmitter group was obtained with their IB magnitude setting at 2.0 mA. Fig. 19 shows the results of this measurement, with the X- and Y-axes corresponding to a spatial arrangement of common-control groups. This spatial arrangement is a horizontally mirrored version of the one shown on the right-hand side of Fig. 2. The average VCSEL output power was obtained by dividing the total measured power by the number of operative VCSELs for each common-control group. These results, which are subject to the same experimental uncertainty described earlier, suggest that neighboring VCSELs tend to have much greater parametric uniformity as compared with those in disparate parts of the array. For the different common-control transmitter groups within a digital-functional channel, variation in the average VCSEL output power ranged from  $\pm 9.7\%$ (ffChan and dgChan) to  $\pm 11.7\%$  (ecChan). Across the entire VCSEL array, variation in the average VCSEL output power was  $\pm 51.1\%$ . A likely reason for the large array-scale variation is the variation in temperature across the chip caused by the different amounts of power dissipated by each functional channel, which could exaggerate the temperature-induced disparity in VCSEL properties from one transmitter circuit group to another. The packaging limitations of the validation PCB was another reason, as only the outermost row of power pads for the transmitter array were wirebonded to, resulting in unwanted IR voltage drops on the power supply and ground rails.

An experiment involving a common-control transmitter group (the same for which the data in Fig. 18 is plotted) on one validation board and a common-control receiver group on another validation board (in the ecChan) was performed to investigate the receiver switching characteristics under dc conditions. The two common-control groups were imaged onto each other using a 4f bulk-lens optical system. The  $nV_{tl}$  and  $nV_{\rm tr}$  inputs for the common-control receiver group were set to complementary states, injecting approximately 60  $\mu$ A of current into one input of each receiver, forcing all receivers to a Logic 0 state. Light from the common-control transmitter group was incident onto corresponding PDs such that the resultant photocurrent was injected into the other input of each receiver, attempting to force them to the Logic 1 state. The IB magnitude of the common-control transmitter group was kept at 0 mA, and the IM magnitude was swept through all possible states between 0 and 4.96 mA. For each IM setting, the average transmitted optical power per VCSEL, estimated received optical power per receiver, and the percentage of



Fig. 20. Receiver switching experiment. All receivers in a common-control group forced to Logic 0 state and illuminated by light from transmitters in a common-control transmitter group from another chip to force them to the Logic 1 state. The percentage of switched receivers is plotted versus estimated received power.



Fig. 21. Transmitter eye diagrams at 250 Mb/s. Vertical scales are 10 mV/div. Eye diagrams along a row correspond to complementary transmitter outputs.

receivers in the common-control group that switched to the Logic 1 state were recorded. Fig. 20 shows the results of this analysis. All but two receivers in the common-control group could eventually be made to switch in this experiment, one of which was verified to be in a permanent stuck-at-zero state using the  $nV_{t1}$  and  $nV_{tr}$  test inputs. It should be noted that this experiment lumps together numerous effects that could detrimentally affect the percentage of switched receivers, such as nonuniform transmitted optical power, power throughput variations in the imaging system across the field of view due to



Fig. 22. Transmitter eye diagrams at data rates of (a) 250, (b) 300, (c) 600, and (d) 900 Mb/s. Vertical scales are 10 mV/div.



Fig. 23. Receiver eye diagrams obtained by probing the electrical output bus at data rates of (a) 10, (b) 25, (c) 100, and (d) 250 Mb/s. Vertical scales are 10 mV/div.



Fig. 24. Optical–electrical–optical eye diagrams of a receiver-transmitter link at data rates of (a) 50, (b) 100, (c) 150, and (d) 250 Mb/s. Vertical scales are 5 mV/div.

misalignment and aberrations, and nonuniformity in R across the common-control receiver group. Fig. 20 indicates that, if provided with enough optical power (approximately 360  $\mu$ W), most receivers could be made to switch notwithstanding these detrimental effects.

The operational performance of the receiver and transmitter circuits was characterized at various data rates and operating conditions. The transmitter circuits in the *dgChan* were tested at high speed by configuring the channel to generate  $2^{16} - 1$  PRBS data and using a high-speed clock signal brought onto the chip via the high-speed connector on the validation PCB. The optical output patterns from individual VCSELs were spatially filtered and captured by an external detector, and eye diagrams were generated on a communication signal analyzer (CSA). Fig. 21 shows eye diagrams from various transmitters at a data rate of



Fig. 25. Calculated relative BER for several single-ended transmitter outputs within a common-control group of the *dgChan* at 250 Mb/s. The spatial arrangement of columns matches the physical location of the corresponding transmitters and the gridlines demarcate individual transmitters in the common-control group.

250 Mb/s. Eye diagrams along a row correspond to the complementary outputs of the same transmitter and exhibit varying degrees of similarity from one transmitter to another. Fig. 22 shows the eye diagrams from one output of a transmitter at data rates of 250, 300, 600, and 900 Mb/s. Despite the long on-chip interconnect between the digital section of the chip and the transmitter array, it is clear that the transmitters perform well at data rates well above their target 250-Mb/s data rate.

The receiver circuits in the *ftChan* were tested by configuring the channel to operate in the *drop* mode and using single-ended PRBS optical data from an external laser source. The corresponding bit on the electrical output bus was accessed via microprobing, and eye diagrams were generated on the CSA. Fig. 23 shows eye diagrams from selected receivers at data rates of 10, 25, 100, and 250 Mb/s. There is a significant amount of switching noise present, as well as voltage overshoot and undershoot that has been intentionally cropped to show the full eyes. A likely cause of the switching noise and voltage overshoot and undershoot is that there were an insufficient number of digital I/O power pad connections made to the chip from the validation PCB. Only two such connections were made for the entire electrical I/O bus.

A similar setup was used to test the combined optical-electrical-optical performance of the receiver and transmitter circuits by operating the ftChan in feed-through mode. The optical input to the receiver was converted to an electrical signal and passed through the digital section of the chip directly to the corresponding transmitter in the transmitter array, which retransmitted the data optically. The optical output from the transmitter was spatially filtered and captured by an external detector, and the CSA was again used to generate eye diagrams. Fig. 24 shows eye diagrams at data rates of 50, 100, 150, and 250 Mb/s. The quality of the eye diagrams in Figs. 23 and 24 both degrade at data rates approaching the target 250-Mb/s data rate. The results of Fig. 24 clearly indicate that the performance of the electrical output pads is the limitation for the results of Fig. 23 and is the source of the switching noise and voltage overshoot and undershoot. Also, it is clear based on the transmitter results from Fig. 22 that the performance of the receiver is the limitation in the receiver-transmitter link, likely due to the long on-chip interconnect that the receiver must drive at its output.

It was not possible to use conventional BER testing equipment to obtain BER data from any of the ODLs due mainly to the packaging limitations of the validation PCB and the data format requirements (rail-to-rail CMOS signals) of the chip I/O pads. Additionally, the error counter of a BER test setup could not be used in conjunction with the *dgChan*—despite its  $2^{16} - 1$  runlength PRBS data generation capability-because synchronization to the PRBS data could not be achieved. Consequently, BER could only be estimated theoretically through manual calculation [49] based on the experimentally obtained eye diagrams. However, BER results obtained in this manner are highly subjective and suspect in terms of absolute accuracy. Additionally, in the case of the transmitters, they are not representative of actual BER performance because only eye diagrams for singleended transmitter outputs were available for analysis. To measure the true BER performance of the transmitter would have required analysis to be performed on the differential output, which could not be obtained experimentally. Nevertheless, such BER data provides meaningful insight into the statistical and relative behavior of individual ODLs within the receiver and transmitter arrays. Fig. 25 shows the calculated relative BER for several single-ended transmitter outputs within a common-control group of the dgChan at a data rate of 250 Mb/s. The data is normalized such that the BER of the worst-performing transmitter output is equal to one. The spatial arrangement of columns in Fig. 25 matches the physical location of the corresponding transmitters in the common-control group. The gridlines demarcate individual transmitters within the common-control group. The relative BERs are all within two orders of magnitude of one another.

#### VII. SUMMARY AND DISCUSSION

We have presented the architecture, circuit design, design verification, and experimental testing of an OE-VLSI chip with a 540-element receiver and transmitter array that employs differential optical signaling and a fully differential electrical architecture. Multiple digital functions suitable for OE-VLSI technology were implemented, and the receiver, transmitter, and digital circuits were designed for robustness, testability, and ease of operability. The receiver and transmitter circuits were designed to specifically meet the performance requirements of the digital circuitry and the electrical interface.

The experimental results presented on the electrical and optical performance of the chip verify the full functionality of the chip design and indicate that the receivers and the digital circuitry were operational close to their target data rates, and that the transmitters exceed their performance target.

When integrated into the chip-to-chip demonstrator system, the system will be capable of an aggregate interchip data bandwidth of [2 chips]  $\times$  [2 arrays/chip]  $\times$  [3  $\times$  128 (ffChan, dgChan, ftChan) + 72 (ecChan) channels/array]  $\times$ [250 Mb/s/channel] = 456 Gb/s. Neglecting the performance of the digital circuitry, the aggregate data bandwidth that can be achieved by the ASIC is limited by the receiver design constraints. The 125- $\mu$ m pitch in the 34 × 35 OED arrays places a severe burden on the receiver, which consequently must drive long electrical interconnects out of the receiver array. Increased aggregate data bandwidths could be readily achieved if the OED pitch were, for example, halved to 62.5  $\mu$ m. One approach would leave the receiver and transmitter designs unchanged (both the receiver and transmitter circuit layouts could fit without modification into the smaller area) and increase the size of the receiver and transmitter arrays four-fold. Another approach would be to redesign the line driver stage of the receiver, which would have an interconnect to drive that is only half as long, for higher performance.

# ACKNOWLEDGMENT

The authors would like to thank A. Chuah and D. Gross for their design efforts, J.-P. Thibodeau for his packaging expertise, and J. Quinn and R. Stevenson of CMC and J. Li of Fibics Inc. for their assistance with ASIC design verification.

#### REFERENCES

- The National Technology Roadmap for Semiconductors—Technology Needs, Semiconductor Industry Association, 1997.
- [2] N. M. Jokerst and A. L. Lentine, Eds., IEEE J. Select. Topics Quantum Electron., Apr. 1996, vol. 2, pp. 3–148.
- [3] N. M. Jokerst, R. Beats, and K. Kobayashi, Eds., IEEE J. Select. Topics Quantum Electron., Mar./Apr. 1999, vol. 5, pp. 143–398.
- [4] Y. Li, E. Towe, and M. Haney, Eds., Proc. IEEE, June 2000, vol. 88, pp. 723–878.
- [5] A. V. Krishnamoorthy and D. A. B. Miller, "Scaling optoelectronic-VLSI circuits into the 21st century: A technology roadmap," *IEEE J. Select. Topics Quantum Electron.*, vol. 2, pp. 55–76, Apr. 1996.
- [6] D. Mathine, "The integration of III-V optoelectronics with silicon circuitry," *IEEE J. Select. Topics Quantum Electron.*, vol. 3, pp. 952–959, June 1997.
- [7] A. V. Krishnamoorthy and K. W. Goossen, "Optoelectronic-VLSI: Photonics integrated with VLSI circuits," *IEEE J. Select. Topics Quantum Electron.*, vol. 4, pp. 899–912, Nov./Dec. 1998.

- [8] A. V. Krishnamoorthy, L. M. F. Chirovsky, W. S. Hobson, R. E. Leibenguth, S. P. Hui, G. J. Zydzik, K. W. Goossen, J. D. Wynn, B. J. Tseng, J. Lopata, J. A. Walker, J. E. Cunningham, and L. A. D'Asaro, "Vertical cavity surface emitting lasers flip-chip bonded to gigabit/s CMOS circuits," *IEEE Photon. Technol. Lett.*, vol. 11, pp. 128–130, Jan. 1999.
- [9] D. V. Plant, M. B. Venditti, E. Laprise, J. Faucher, K. Razavi, M. Chateauneuf, A. G. Kirk, and J. S. Ahearn, "256 channel bidirectional optical interconnect using VCSELs and photodiodes on CMOS," *J. Lightwave Technol.*, vol. 19, pp. 1093–1103, Aug. 2001.
- [10] A. C. Walker, M. P. Y. Desmulliez, M. G. Forbes, S. J. Fancey, G. S. Butler, M. R. Taghizadeh, J. A. B. Dines, C. R. Stanley, G. Pennelli, A. R. Boyd, P. Horan, D. Byrne, J. Hegarty, S. Eitel, H.-P. Gauggel, K.-H. Gulden, A. Gauthier, P. Benabes, J.-L. Gutzwiller, and M. Goetz, "Design and construction of an optoelectronic crossbar switch containing a terabit per second free-space optical interconnect," *IEEE J. Select. Topics Quantum Electron.*, vol. 5, pp. 236–249, Mar./Apr. 1999.
- [11] A. L. Lentine, K. W. Goossen, J. A. Walker, J. E. Cunningham, W. Y. Jan, T. K. Woodward, A. V. Krishnamoorthy, B. J. Tseng, S. P. Hui, R. E. Leibenguth, L. M. F. Chirovsky, R. A. Novotny, D. B. Buchholz, and R. L. Morrison, "Optoelectronic VLSI switching chip with greater than 1 Tbit/s potential optical I/O bandwidth," *Electron. Lett.*, vol. 33, pp. 894–895, May 1997.
- [12] A. L. Lentine, K. W. Goossen, J. A. Walker, L. M. F. Chirovsky, L. A. D'Asaro, S. P. Hui, B. J. Tseng, R. E. Leibenguth, J. E. Cunningham, W. Y. Jan, J.-M. Kuo, D. W. Dahringer, D. P. Kossives, D. D. Bacon, G. Livescu, R. L. Morrison, R. A. Novotny, and D. B. Buchholz, "High-speed optoelectronic VLSI switching chip with >4000 optical I/O based on flip-chip bonding of MQW modulators and detectors to silicon CMOS," *IEEE J. Select. Topics Quantum Electron.*, vol. 2, pp. 77–84, Apr. 1996.
- [13] A. V. Krishnamoorthy, J. E. Ford, F. E. Kiamilev, R. G. Rozier, S. Hunsche, K. W. Goossen, B. Tseng, J. A. Walker, J. E. Cunningham, W. Y. Jan, and M. C. Nuss, "The AMOEBA switch: An optoelectronic switch for multiprocessor networking using dense-WDM," *IEEE J. Select Topics Quantum Electron.*, vol. 5, pp. 261–275, Mar./Apr. 1999.
- [14] A. G. Kirk, D. V. Plant, T. H. Szymanski, Z. G. Vranesic, F. A. P. Tooley, D. R. Rolston, M. H. Ayliffe, F. K. Lacroix, B. Robertson, E. Bernier, and D. F. Brosseau, "Design and implementation of a modulator-based freespace optical backplane for multiprocessor applications," *Appl. Opt.*, 2003, to be published.
- [15] C.-S. Li, H. S. Stone, Y. Kwark, and C. M. Olsen, "Fully differential optical interconnections for high-speed digital systems," *IEEE Trans. VLSI Syst.*, vol. 1, pp. 151–163, June 1993.
- [16] M. Govindarajan, S. Siala, and R. N. Nottenburg, "Optical receiver systems for high speed parallel digital data links," *J. Lightwave Technol.*, vol. 13, pp. 1555–1565, July 1995.
- [17] M. B. Venditti and D. V. Plant, "Receiver operational yield in optoelectronic-VLSI applications," *Proc. SPIE*, vol. 4788, pp. 39–49, 2002.
- [18] —, "On the design of large receiver and transmitter arrays for OE-VLSI applications," J. Lightwave Technol., submitted for publication.
- [19] M. B. Venditti, J. Faucher, and D. V. Plant, "Switching noise considerations in the design of OE-VLSI transmitter and receiver circuits," in *Proc. Opt. Computing Conf.*, Apr. 2002, pp. 267–269.
- [20] D. R. Rolston, D. V. Plant, T. H. Szymanski, H. S. Hinton, W. S. Hsiao, M. H. Ayliffe, D. Kabal, M. B. Venditti, A. V. Krishnamoorthy, K. W. Goossen, J. A. Walker, B. Tseng, S. P. Hui, J. C. Cunningham, and W. Y. Jan, "A hybrid-SEED smart pixel array for a four-stage intelligent optical backplane demonstrator," *IEEE J. Select. Topics Quantum Electron.*, vol. 2, pp. 97–105, Apr. 1996.
- [21] M. W. Haney, M. P. Christensen, P. Milojkovic, G. J. Fokken, M. Vickberg, B. K. Gilbert, J. Rieve, J. Ekman, P. Chandramani, and F. E. Kiamilev, "Description and evaluation of the *FAST-Net* smart pixel-based optical interconnection prototype," *Proc. IEEE*, vol. 88, pp. 819–828, June 2000.
- [22] C. B. Kuznia, J.-M. Wu, C.-H. Chen, B. Hoanca, L. Cheng, A. G. Weber, and A. A. Sawchuk, "Two-dimensional parallel pipeline smart pixel array cellular logic (SPARCL) processors-chip design and system implementation," *IEEE J. Select. Topics Quantum Electron.*, vol. 5, pp. 376–386, Mar./Apr. 1999.
- [23] D. Fey, W. Erhard, M. Gruber, J. Jahns, H. Bartlet, G. Grimm, L. Hoppe, and S. Sinzinger, "Optical interconnects for neural and reconfigurable VLSI architectures," *Proc. IEEE*, vol. 88, pp. 838–848, June 2000.

- [24] N. McArdle, M. Naruse, and M. Ishikawa, "Optoelectronic parallel computing using optically interconnected pipelined processing arrays," *IEEE J. Select. Topics Quantum Electron.*, vol. 5, pp. 250–260, Mar./Apr. 1999.
- [25] F. E. Kiamilev and R. G. Rozier, "Smart pixel IC layout methodology and its application to photonic page buffers," *Int. J. Optoelectron.*, vol. 11, pp. 199–215, 1997.
- [26] R. G. Rozier, F. E. Kiamilev, and A. V. Krishnamoorthy, "Design and evaluation of a photonic FFT processor," *J. Parallel Distrib. Comput.*, vol. 41, pp. 131–136, 1997.
- [27] IEEE Standard Hardware Description Language Based on the Verilog<sup>®</sup> Hardware Description Language, IEEE Stand. 1364, 1995.
- [28] F. A. P. Tooley, "Challenges in optically interconnecting electronics," IEEE J. Select. Topics Quantum Electron., vol. 2, pp. 3–13, Apr. 1996.
- [29] F. E. Kaimilev, J. Ekman, P. Chandramani, A. V. Krishnamoorthy, and R. G. Rozier, "CMOS DSP and microprocessor cores using optoelectronic VLSI," in *Proc. Southwest Symp. Mixed-Signal Design*, 2000, pp. 9–13.
- [30] F. E. Kiamilev and A. V. Krishnamoorthy, "A high-speed 32-channel CMOS VCSEL driver with built-in self-test and clock generation circuitry," *IEEE J. Select. Topics Quantum Electron.*, vol. 5, pp. 287–295, Mar./Apr. 1999.
- [31] K. Azadet, E. F. Haratsch, H. Kim, F. Saibi, J. H. Saunders, M. Shaffer, L. Song, and M.-L. Yu, "Equalization and FEC techniques for optical transceivers," *IEEE J. Solid-State Circuits*, vol. 37, pp. 317–327, Mar. 2002.
- [32] J. Faucher, M. B. Venditti, and D. V. Plant, "Application of parallel forward error correction in two-dimensional optical data links," *J. Lightwave Technol.*, vol. 21, pp. 466–475, Feb. 2003.
- [33] M. A. Neifeld and S. K. Sridharan, "Parallel error correction using spectral Reed-Solomon codes," *J. Opt. Commun.*, vol. 18, pp. 1144–150, 1997.
- [34] T. K. Matsushima, T. Matsushima, and S. Hirasawa, "Parallel architecture for high-speed Reed-Solomon codec," in *Proc. SBT/IEEE Int. Telecommun. Symp.*, vol. 2, 1998, pp. 468–473.
- [35] M. J. E. Golay, "Notes on digital coding," Proc. IRE, vol. 37, p. 657, 1949.
- [36] W. Cao, "High-speed parallel VLSI-architecture for the (24,12) golay decoder with optimized permutation decoding," in *Proc. IEEE Int. Symp. Circuit Systems Conf.*, vol. 4, 1996, pp. 61–64.
- [37] A. V. Krishnamoorthy, K. W. Goossen, L. M. F. Chirovsky, R. G. Rozier, P. Chandramani, S. P. Hui, J. Lopata, J. A. Walker, and L. A. D'Asaro, "16 × 16 VCSEL array flip-chip bonded to CMOS VLSI circuit," *IEEE Photon. Technol. Lett.*, vol. 12, pp. 1073–1075, Aug. 2000.
- [38] P. Bontoux, F. Mieyeville, I. O'Connor, F. Gaffiot, and G. Jacqumod, "Design and optimization of optical links based on VHDL-AMS modeling," in *Proc. IEEE/ACM Int. Workshop Behavioral Modeling and Simulation*, 2000, pp. 62–67.
- [39] F. Gaffiot, K. H. K. Vuorinen, F. Mieyeville, I. O'Connor, and G. Jacqumod, "Behavioral modeling for hierarchical simulation of optronic systems," *IEEE Trans. Circuits Syst. II*, vol. 46, pp. 1316–1322, Oct. 1999.
- [40] F. Mieyeville, G. Jacquemod, F. Gaffiot, and M. Belleville, "A behavioral opto-electro-thermal VCSEL model for simulation of optical links," *Sens. Actuators A, Phys.*, vol. 88, pp. 209–219, 2001.
- [41] P. M. VanPetegham and J. F. Duque-Carillo, "A general description of common-mode feedback in fully-differential amplifiers," in *Proc. IEEE Int. Symp. Circuits Syst.*, vol. 4, 1990, pp. 3209–3212.
- [42] J. A. Tabler, "An integrated adaptive bias solution for zero passive component count high-performance mixed-signal IC's," Ph.D. dissertation, Georgia Inst. Technology, Atlanta, GA, 1997.
- [43] M. Bazes, "Two novel fully complementary self-biased CMOS differential amplifiers," *IEEE J. Solid-State Circuits*, vol. 26, pp. 165–168, Feb. 1991.
- [44] R. J. Baker, H. W. Li, and D. E. Boyce, CMOS: Circuit Design, Layout, and Simulation. Piscataway, NJ: IEEE Press, 1998.
- [45] A. V. Krishnamoorthy et al., "Triggered receivers for optoelectronic VLSI," Electron. Lett., vol. 36, pp. 249–250, Feb. 2000.
- [46] N. H. Weste and K. Eshraghian, *Principles of CMOS VLSI Design*, 2nd ed. Reading, MA: Addison-Wesley, 1993.
- [47] P. H. Bardell, W. H. McAnney, and J. Savir, *Built-In Test for VLSI: Pseudorandom Techniques*. New York: Wiley, 1987.
- [48] F. E. Kiamilev, J. S. Lambirth, R. G. Rozier, and A. V. Krishnamoorthy, "Design of a 64-bit, 100 mips microprocessor core IC for hybrid CMOS-SEED technology," in *Proc. IEEE Workshop Massively Parallel Processing Using Opt. Interconnects*, Oct. 1996, pp. 53–60.
- [49] G. Keiser, Optical Fiber Communication, 3rd ed. New York: McGraw-Hill, 2000.



Michael B. Venditti (S'94–A'96) received the B.Eng. (honors), M.Eng., and Ph.D. degrees in electrical and computer engineering from McGill University, Montreal, QC, Canada, in 1996, 1999, and 2003, respectively. His research focused on the design of optical receivers and transmitters, and ASIC design for optoelectronic-VLSI applications.

In 1997 and 1999, he was involved in four-month internships at Nortel Networks and Sanders, a Lockheed Martin company (now BAE Systems/TeraConnect Inc.), respectively. In 2000, he was a

Faculty Lecturer for the Department of Electrical and Computer Engineering at McGill University. He now works at PMC-Sierra in Montreal, QC, Canada, in the mixed-signal design group. His research interests include OEDs, free-space optical interconnects, mixed-signal ASIC design, and applications of optoelectronic-VLSI technology.

Dr. Venditti is a member of the IEEE Lasers and Electro-Optics, Solid-State Circuits, and Circuits and Systems Societies. He has received undergraduate and graduate scholarships from the Natural Sciences and Engineering Research Council of Canada and from Le Fonds pour la Formation de Chercheurs et l'Aide à la Recherche du Québec. He was recently awarded a 2002 LEOS Graduate Student Fellowship.



**Emmanuelle Laprise** received the B.Eng. degree in computer engineering in 1996 and the M.Eng. degree in electrical engineering in 2001 from McGill University, Montreal, QC, Canada. Her master's research focused on applying standard VLSI design and testing techniques to ASICs that incorporated optical IOs.

A significant portion of the work for this paper was done while she was working as a Research Assistant for Prof. Plant in the Photonics Laboratory,

McGill University. She is currently working as a ju-

nior product design engineer for PMC-Sierra, Montreal, QC, Canada.



**Julien Faucher** (S'99–M'99) received the B.Eng. and M.Eng. degrees in electrical and computer engineering from McGill University, Montreal, QC, Canada, in 1999 and 2001, respectively. He is currently pursuing the Ph.D. degree at McGill University, focusing on the design of optical receivers and ASIC design for optical networks.

Mr. Faucher is a member of the IEEE Lasers and Electro-Optics Society and the IEEE Solid-State Circuits Society. He has received graduate scholarships from the Natural Sciences and Engineering Research

Council of Canada and from Le Fonds pour la Formation de Chercheurs et l'Aide à la Recherche du Québec. He was recently awarded a Richard H. Tomlinson Doctoral Fellowship from McGill University.



**Pierre-Olivier Laprise** received the B.Eng. degree in computer engineering from McGill University, Montreal, QC, Canada, in 2001. He is currently pursuing the M.Eng. degree at McGill University, focusing on reconfigurable computing and its applications to video processing and multiple camera cooperation.

In 2000 and 2001, he worked for McGill University's Photonic Systems Group while completing his B.Eng. His research interests include design of real-time reconfigurable FPGA systems, reconfig-

urable computing, and parallel processing.



**J. Eduardo A. Lugo** received the Ph.D. degree in physics from the Morelos State University, Cuernavaca, Mor., Mexico, in 1998. His research involved experimental and theoretical studies of porous silicon nanostructures.

In 1998, he was at the Department of Physics of the Morelos State University, where his research focused on porous silicon multilayers and molecular dynamics simulations. From 1999 to 2000, he was a Postdoctoral Fellow in the Electrical and Computer Engineering Department and the Center for Future

Health at the University of Rochester, Rochester, NY, working with biosensors and photonic bandgap materials based on nanostructures of silicon. In 2000, he was at the Center for Energy Research of the National University of Mexico (UNAM), where his work entailed research on the electro-optical properties of nanostructures and multilayer filters. Since 2001, he has been with the Photonic Systems Group in the Electrical and Computer Engineering Department at McGill University, Montreal, QC, Canada, working with OE-VLSI systems. Currently, he is interested in developing new VCSELs and tunable photonics bandgap devices.



**David V. Plant** (S'86–M'89) received the Ph.D. degree in electrical engineering from Brown University, Providence, RI, in 1989.

From 1989 to 1993, he was a Research Engineer in the Department of Electrical and Computer Engineering at the University of California, Los Angeles (UCLA). He has been a Professor and Member of the Photonic Systems Group within the Department of Electrical and Computer Engineering at McGill University, Montreal, QC, Canada, since 1993. Since joining McGill University, He has been studying op-

tical interconnects for very short reach (VSR) applications. He is the Major Project Leader of the Photonic Interconnection Systems major project within the Canadian Institute for Telecommunications Research (CITR) and a project co-leader within the Canadian Institute for Photonic Innovation (CIPI). From 2000 to 2001, he took a leave of absence from McGill University to become the Director of Optical Integration at Accelight Networks, a start-up company based in Pittsburgh, PA and Ottawa, ON, Canada, which will deploy optical internetworking equipment.

Dr. Plant received the Outstanding Departmental Teaching Award and the Faculty of Engineering Teaching Award from McGill University in 1996. In recognition of his accomplishments in research and teaching, he was named a James McGill Professor in 2001.